Author’s response to reviews

Title: A Rare Case of Spontaneous Hepatic Rupture in a Pregnant Woman

Authors:
Xiao Zhou (nj20zx@163.com)
Meng Zhang (zhangmengmmz@163.com)
Zhuang Liu (kewlnlzh@gmail.com)
Meili Duan (beauty9659@hotmail.com)
Lei Dong (wojiushidonglei@126.com;18618334341@163.com)

Version: 5 Date: 04 Feb 2018

Author’s response to reviews:

List of Responses

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “A Rare Case of Spontaneous Hepatic Rupture in a Pregnant Woman” (ID: PRCH-D-17-00765R2). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the technical comments:

1. Response to comment: (revise the figure files)
Response: We have made correction according to the technical comments.

2. Response to comment: (revise the manuscript as a single, final, clean version)
Response: We have made correction according to the technical comments.
Special thanks to you for your good comments.

Responds to the Editor Comments:( view the reports of the reviewers via the submission system)
Response: We have studied comments carefully and have made corrections.
Special thanks to you for your good comments.

We tried our best to improve the manuscript and made some changes in the manuscript. These changes will not influence the content and framework of the paper.
We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.

Yours sincerely,
Xiao Zhou

Corresponding author:
Name: Lei Dong
E-mail: 18618334341@163.com